Author's response to reviews

Title: A tailored, interactive health communication application for patients with type 2 diabetes: study protocol of a randomised controlled trial

Authors:

Nina Weymann (n.weymann@uke.de)
Martin Haerter (m.haerter@uke.de)
Joerg Dirmaier (dirmaier@uke.de)

Version: 2 Date: 4 February 2013

Author's response to reviews: see over
Dear Ladies and Gentlemen,

Thank you for your feedback on our study protocol.

I am pleased to send you the ethics approval by the Ethics Committee of the Medical Council of Hamburg and the grant notice informing us about public funding by the Federal Ministry of Education and Science.

There are no manuscripts under submission with other journals based on this study protocol. The other manuscripts mentioned in our first cover letter are in preparation and will be submitted in the next weeks. We are working on publishing all relevant steps of our study (study protocol, needs assessment, system development, RCT) in order to make them transparent for other researchers planning a similar trial.

I would appreciate if you considered our study protocol for publication.

Sincerely,

Nina Weymann
Sincerely,

Prof. Dr. Dr. Martin Härter